Author's response to reviews

Title: DCE-MRI assessment of the effect of Epstein-Barr Virus-encoded latent membrane protein-1 targeted DNAzyme on tumor vasculature in patients with nasopharyngeal carcinomas

Authors:

Wei-Hua Liao (ouwenliao@163.com)
Li-Fang Yang (yanglifang99@hotmail.com)
Xiao-Yu Liu (lxy9827@qq.com)
Gao-Feng Zhou (peak.zh@163.com)
Wu-Zhong Jiang (13973126486@189.cn)
Bob-Lei Hou (bhou@hsc.wvu.edu)
Lun-Quan Sun (lunquansun@csu.edu.cn)
Ya Cao (ycao98@vip.sina.com)
Xiao-Yi Wang (185223273@qq.com)

Version: 5 Date: 14 October 2014

Author's response to reviews: see over
Dear editorial

Thank you for your valuable feedback. We have provided point-by-point responses (in BOLD type) to each comments, and the page and line numbers provided refer to our revised manuscript submitted Oct 14, 2014.

1. Under Protocol design, include a sentence saying why the two randomized group were not equal in size.
   **Response:** We have added the sentence to explain why the randomization of patients was not ideal, please see P7 Line 182-183.

2. First sentence in Results, exclude the word further.
   **Response:** We have deleted this word. See P9 Line 262.

3. In Figures 2 and 3, draw tumour outline or insert arrows pointing to the tumour in the two upper rows of images.
   **Response:** We have inserted arrows to the tumour in the two upper rows of the two figures. See the revised Fig. 2 and 3.

   Thanks for the thoughtful review. If you still have any further queries, do not hesitate to inform us and we shall make corrections accordingly.